Author's response to reviews

Title: Socio-economic inequalities in curative health-seeking for children in Egypt: Analysis of the 2008 Demographic and Health Survey

Authors:

Lenka Benova (Lenka.Benova@lshtm.ac.uk)
Oona MR Campbell (Oona.Campbell@lshtm.ac.uk)
George B Ploubidis (G.Ploubidis@ioe.ac.uk)

Version: 3
Date: 10 October 2015

Author's response to reviews: see over
Response to editor

Manuscript number: 1205836889175908

Dear Professor Arah,

We would like to express our appreciation for the opportunity to revise our manuscript.

We have revised the Authors’ Contributions sections to specifically mention the contributions of Professor Oona Campbell. If there are any other questions or clarifications that you require, please do not hesitate to contact me.

Yours Sincerely,

Lenka Benova, on behalf of the author team
Response

Editor

Comment 1: We would like to ask you to please include Oona MR Campbell's contributions specifically in your Author's contributions section.

Response: We thank the editor for this request. We have revised the wording of this section to state:

“LB and GBP conceptualised the study. LB conducted the analysis and prepared the first draft of the manuscript. LB, OMRC and GBP contributed to the interpretation of the data and the preparation of the final manuscript.”